\documentstyle[12pt,twoside]{article}
\input{macros-course}
\begin{document}


\vspace{1in}

\begin{center}
{\bf As I am out of town, my Monday 3:00 and 4:00 tutorials are being
rescheduled. You may choose any one of the following to attend. -Brian}
\end{center}

\vspace{1in}

\begin{center}

\begin{tabular}{|c||c|ll|}
\hline
& TA & \multicolumn{2}{c|}{Tutorial (Monday)}\\
\hline
1 &  {\bf Steve }     	&  3:00, & 24--316 \\
2 &            	&  4:00, & 24--316 \\
\hline
3 &  {\bf Ravi }     	&  3:00, & 34--498 \\
\hline
&  & \multicolumn{2}{c|}{ (Tuesday)}\\
4 &  {\bf Brian } &  4:00, & 2--588 \\
5 &           	&  5:00, & 2--588 \\
\hline
\end{tabular}

\end{center}




\end{document}
